

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hener, Timo; Rainer, Helmut; Bauernschuster, Stefan

Conference Paper When Labor Disputes Bring Cities to a Standstill: The Impact of Public Transit Strikes on Traffic, Accidents, Air Pollution, and Health

Beiträge zur Jahrestagung des Vereins für Socialpolitik 2015: Ökonomische Entwicklung -Theorie und Politik - Session: Labor 5, No. E17-V2

Provided in Cooperation with:

Verein für Socialpolitik / German Economic Association

Suggested Citation: Hener, Timo; Rainer, Helmut; Bauernschuster, Stefan (2015) : When Labor Disputes Bring Cities to a Standstill: The Impact of Public Transit Strikes on Traffic, Accidents, Air Pollution, and Health, Beiträge zur Jahrestagung des Vereins für Socialpolitik 2015: Ökonomische Entwicklung - Theorie und Politik - Session: Labor 5, No. E17-V2, ZBW - Deutsche Zentralbibliothek für Wirtschaftswissenschaften, Leibniz-Informationszentrum Wirtschaft

This Version is available at: https://hdl.handle.net/10419/112957

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

When Labor Disputes Bring Cities to a Standstill: The Impact of Public Transit Strikes on Traffic, Accidents, Air Pollution, and Health

Stefan Bauernschuster University of Passau CESifo Ifo Institute Timo Hener Ifo Institute

University of Munich CESifo Ifo Institute

Helmut Rainer

June 2015

Abstract

Many governments have banned strikes in public transportation. Whether this can be justified depends on whether strikes endanger public safety or health. We use time-series and cross-sectional variation in powerful registry data to quantify the effects of public transit strikes on urban populations in Germany. Due to higher traffic volumes and longer travel times, total car hours operated increase by 11% to 13% during strikes. This effect is accompanied by a 14% increase in vehicle crashes, a 20% increase in accident-related injuries, a 14% increase in particle pollution, and an 11% increase in hospital admissions for respiratory diseases among young children.

JEL codes: L91, R41, Q53, I18

Email addresses: stefan.bauernschuster@uni-passau.de (Stefan Bauernschuster), hener@ifo.de (Timo Hener), rainer@ifo.de (Helmut Rainer).

"The right to strike may be restricted or prohibited [...] in essential services [...] (that is, services the interruption of which would endanger the life, personal safety or health of the whole or part of the population). The following do[es] not constitute [an] essential service[...]: transport generally." – International Labour Organization (2006, para. 576 and para. 587)

"Many public services are considered essential: police officers and firefighters, for example. Strikes are prohibited for this very reason. They are critical for the public on a day-to-day basis. The reliability of public transit should be no different."

- Robert S. Huff, California State Senate Republican Leader (January 13, 2014)

1. Introduction

In 1951, the International Labour Organization (ILO) set up the Committee on Freedom of Association (CFA). Shortly after its inception, the CFA declared strike action to be a fundamental right of organized labor (Gernigon *et al.*, 1998; Gross, 1999). Yet, where workers providing essential public services are concerned, the right to strike is often limited or even denied by national laws or regulations. The most common restriction is a ban on strikes by armed forces, policemen and firefighters, for the legitimate reason that those walkouts would endanger the life, personal safety or health of the whole or parts of the population.¹ But is that true of strikes by public transit workers? Two extreme positions shape answers to this question. According to the ILO, public transportation does not constitute an essential public service (ILO, 2006, para. 587). Thus, some commentators argue that strikes by transit workers mainly pose an economic threat, which—being the very essence of industrial action—does not justify a strike ban (Swearengen, 2010). Policy-makers, by contrast, commonly regard mass transit as an essential public service, which segues into the wider concern that major cities and their inhabitants are highly vulnerable to transit strikes.² This is exemplified by attempts in numerous countries to also exclude transit workers from the right to strike.

New York City's *Taylor Law*, which was put into effect in response to a transit strike in 1966, represents an example of a particularly draconian measure. Under Section 210, the law

¹As the first quote above illustrates, the ILO recognizes that strikes may be restricted or prohibited in essential services, which are defined to include: the hospital sector, electricity services, water supply services, the telephone service, the police and armed forces, the fire-fighting services, public and private prison services, and air traffic control (ILO, 2006, para. 585).

²Thus, the second quote above from a California politician, which was made following a strike by workers of the Bay Area Rapid Transit (BART) system in 2013. The position expressed in this statement received bipartisan support. Indeed, in the aftermath of the same strike, California State Senate Democratic candidate Steve Glazer expressed his "[...] support [for] state legislation to prohibit transportation workers from striking". For more, see http://calwatchdog.com/2014/01/24/democrats-crash-transit-strike-ban/ (accessed December 8, 2014).

prohibits any strike or other concerted stoppage of work or slowdown by public employees (NYS Department of State, 2009). Instead, it prescribes binding arbitration by a state agency to resolve bargaining deadlocks between unions and employers. Violations against the prohibition on strikes are punishable with hefty penalties. The fine for an individual worker is twice the striking employee's salary for each day the strike lasts. In addition, union leaders face imprisonment. Since its inception in 1967, the *Taylor Law* has generated a lot of controversy. To proponents, it was successful in averting several potential transit strikes that would have imposed significant costs on the city and its inhabitants (OECD, 2007a). Indeed, New York City has only seen two transit strikes over the past four decades—in 1980 and in 2005. In both cases, harsh monetary penalties were imposed on workers and unions. The 2005 transit strike additionally led to the imprisonment of a union leader, and saw the Transport Workers Union (TWU) filing a formal complaint with the International Labour Organization (ILO). Since then, the ILO has urged the United States government to restore the right of transit workers to strike, arguing that they do not provide essential services justifying a strike ban (ILO, 2011, p. 775). So far, the *Taylor Law* has not been amended in this direction.

This paper aims to answer two questions that are at the heart of the *Taylor Law* controversy and similar debates elsewhere: Do strikes in the public transportation sector cause disruptions that endanger the safety and health of urban populations? And how large are the costs of transit strikes to non-involved third parties? To get at these questions, our analysis uses time series and cross-sectional variation in powerful registry data to quantify the effects of public transit strikes in five domains: traffic volumes, travel times, accident risk, pollution emissions, and health (see Figure 1). The context for our study are the five largest cities in Germany, which provides us with an ideal setting. In particular, in contrast to countries that have imposed *de jure* restrictions on public transit strikes, German courts *de facto* protect the right to strike in this sector. As a consequence, Germany regularly faces strikes by transit workers.

Our analysis exploits 71 one-day strikes in public transportation over the period from 2002 to 2011. We identify the daily effects of these strikes using both time series and cross-sectional variation in our data. In a first step, we estimate the impact on the total length of time that cars are in operation (henceforth, total car hours operated). To do so, we make use of two data sources. First, we use hourly information from official traffic monitors to estimate the effect of transit strikes on traffic volumes. Second, we use congestion data based on GPS speed measurements from *TomTom*, a global supplier of navigation and location products and services, to estimate the effect on travel times. Combining the two estimates allows us to compute the effect on total car hours operated. In a second step, we explore likely knock-on consequences by expanding the analysis in three directions. First, we assess the impact of strikes on the incidence



and severity of car accidents using detailed register data which includes all vehicle crashes recorded by the German police. Second, to investigate the effect on atmospheric pollution, we draw on hourly data from official air monitors. Third, we explore the effect on human health using register data which includes information about all patients admitted to all German hospitals. Our identification strategy is based on a generalized difference-in-differences approach. It flexibly captures daytime and day-of-week patterns, seasonality effects, and long-run time trends, which are all allowed to vary by city.

What emerges is a picture of remarkable consistency. During the morning peak of a strike day, total car hours operated increase by 11% to 13%. This increase can be decomposed into two separate effects: a 2.5% to 4.3% increase in the number of cars on roads and a 8.4% increase in travel times. In addition, our results suggest that transit strikes pose a non-negligible threat to public safety and public health. We find a 14% increase in the number of vehicle crashes, which is accompanied by a 20% increase in accident-related personal injuries. Moreover, we observe that transit strikes have sizeable effects on ambient air pollution. Emissions of particulate matter increase by 14%, while nitrogen dioxide concentrations in ambient air increase by 4%. Finally, analyzing health outcomes related to air pollution, we find that young children are subject to negative health effects. Among this subgroup, hospital admissions for respiratory diseases increase by 11% on strike days.

The costs of strikes—both to the parties directly involved in a dispute and to the public at large—have been the subject of extensive research since the mid-20th century. Until the 1990s,

the main conclusion of the literature was that strikes impose significant financial costs on the workers and the firm directly involved in walkouts, but only negligible costs in most cases on non-involved third parties (Kaufmann, 1992). Our study firmly rejects this conclusion: based on our estimates, the increase in aggregate travel time caused by a single strike corresponds to 1,550 full-time equivalent work weeks. This translates into third-party congestion costs of \notin 3.2 million per strike or \notin 228.9 million for all 71 strikes in our sample.

To the best of our knowledge, this study is the first to examine whether strikes in the transportation sector can put public safety and health at risk. There are, however, a few impressive empirical studies of strike impacts in other areas of the public sector. Focusing on the hospital sector, Gruber and Kleiner (2012) investigate the effects of a nurse strike on patient outcomes. After controlling for time and hospital specific heterogeneity, they observe increased mortality and readmission rates and conclude that strikes in hospitals kill. This result contradicts earlier studies that did not as rigorously control for unobserved factors (see, e.g., Cunningham *et al.*, 2008; Pantell and Irwin, 1979). Another study by Mustard et al. (1995) highlights that there are fewer caesarian births during strike periods, which is suggestive of behavioral effects in hospitals. Examining walkouts in the education sector, Belot and Webbink (2010) and Baker (2013) find that teacher strikes had negative effects on student achievement in Belgium and Canada. Finally, there are a few interesting studies of strike impact in the private sector. Krueger and Mas (2004) show that strikes in tire production facilities decreased the quality of tires resulting in an increase of fatal accidents. In a similar vein, Mas (2008) finds that strikes at *Caterpillar* led to lower product quality.

Our paper is also related to a growing literature in economics that examines the role of mass transit in mitigating agglomeration diseconomies such as traffic congestion, accident risk and pollution emissions. In an influential study, Duranton and Turner (2011) coined the notion of the "fundamental law of road congestion". Theoretically, the idea is that the provision of public transit is unlikely to relieve the overall level of congestion in a city since it only results in additional traffic that continues to rise until peak congestion returns to its natural level. The authors provide empirical evidence in support of this mechanism. There are, however, a few notable papers which point in the opposite direction. Anderson (2014) exploits a 35-day strike in 2003 by Los Angeles transit workers to evaluate the net benefits of urban mass transportation. Using a regression-discontinuity design, he estimates the total congestion relief benefit of operating the Los Angeles transit system to lie between \$1.2 billion to \$4.1 billion per year. Nelson *et al.* (2007) provide structural estimates suggesting that the rail transit system in Washington, D.C., generates congestion-reduction benefits that exceed rail subsidies. Finally, Chen and Whalley (2012) quantify the effects of urban rail transit on air quality using the sharp discontinuity in

ridership on the opening day of a new rail transit system in Taipei. Their findings suggest that the opening of the rail transit system caused a 5 to 15 percent reduction in carbon monoxide emissions.³ Our study, which exploits strikes of one day in length or less, contributes to this literature by showing that even short-term disruptions of mass transit services can have far reaching consequences for urban populations in terms of time lost to travel, accident risk, air pollution and health.

The remainder of the paper is organized as follows. Section 2 provides the institutional setting and discusses how transit strikes might affect cities and their inhabitants. Section 3 describes the data. Section 4 outlines the empirical strategy, followed by the results in Section 5. Section 6 discusses the size of the effects by monetizing the third party costs of transit strikes and comparing them to the private costs of struck employers.

2. Background

2.1. The Role of Public Transit and the Regulation of Labor Relations

The five largest German cities, home to roughly 8.2 million people, are characterized by an intensive use of public transportation. In 2013, Berlin, Hamburg, Munich, Cologne and Frankfurt together accounted for a total number of 3.4 billion public transit users in their metropolitan areas.⁴ This corresponds to an average 9.3 million passengers a day. In Berlin, the German capital, roughly 43% of commuters use public transit, while about 38% travel by car (Wingerter, 2014). Public transportation networks are extensive in all sample cities. In Hamburg, for example, the transportation network comprises 91 subway stations, 68 suburban train stations (S-Bahn), more than 1,300 bus stops connecting a network of nearly 1,200 km in a city with less than 2 million inhabitants. The importance of public transportation in major German cities is comparable to the role it plays in the largest city in the United States. New York City has a population of roughly 8.4 million people. In 2014, its Metropolitan Transportation Authority moved about 9 million riders per day or 3.3 billion passengers a year on subways, buses and railroads.⁵ Approximately 56 percent of commuters in New York City use public transit, while

³Relatedly, Lalive *et al.* (2013) analyze a railway reform in Germany which substantially increased the frequency of regional passenger services. Their results suggest that the reform reduced the number of severe road traffic accidents, carbon monoxide, nitrogen monoxide, nitrogen dioxide pollution and infant mortality.

^{41.321} million Berlin http://www.vbb.de/de/article/verkehrsverbund/derpassengers in (see verbund-in-zahlen/12552.html), 855 million in passengers Hamburg (see http://www.hvv.de/pdf/aktuelles/publikationen/hvv_zahlenspiegel_2013.pdf), 663 million passengers in Munich (see http://www.mvv-muenchen.de/de/der-mvv/mvv-in-zahlen/), 277 million passengers in Cologne (see http://www.kvb-koeln.de/newsfiles/310b105c8ee08bf447f1df1f89cd3a87.pdf) and 203 million passengers in Frankfurt (see http://www.traffiq.de/1483.de.presse informationen.html? pi=126798).

⁵See http://www.apta.com/resources/statistics/Documents/Ridership/2014-q2-ridership-APTA.pdf.

about 27 percent travel by car.⁶

While the use of mass transit in New York City and major German cities is comparable, the regulation of labor relations in the public transportation sector differs markedly. As mentioned above, New York City's Taylor Law prohibits strikes by transit workers under the threat of harsh penalties. Other cities in the United States with no-transit-strike laws include Chicago, Boston and Washington, D.C. For a German, it must come as a surprise that many countries impose de *jure* restrictions on strikes in the public transportation sector. Indeed, in Germany, the right to strike is a fundamental right based on the Freedom of Association (Koalitionsfreiheit) as laid out in Article 9(3) of the constitution (Grundgesetz). Only civil servants, judges and soldiers are excluded from the right to strike. Until the 1990s, the big infrastructure industries—i.e., telecommunications, postal and public transportation services—were state monopolies. Workers in these industries had civil servant status and thus were not allowed to strike. However, when these industries were gradually privatized during the 1990s, newly hired workers were no longer given civil servant status and therefore gained the right to strike. Today, public transit workers, whether employed by Germany's rail operator Deutsche Bahn or local public transport providers, are allowed to engage in industrial action. The only de facto restriction on transit workers' right to strike is that the parties of an industrial conflict are responsible for the provision of a minimum service (Klaß et al., 2008). This is intended to act as a balance of their interests with those of non-involved third parties.⁷

In Germany, industrial action by transit workers is typically announced one day ahead of a strike. However, at that time, there is still substantial uncertainty as to exactly which services will be affected and to what degree. Thus, the actual extent of a strike cannot be clearly assessed prior to the start of a strike. Although public transit strikes generally do not shut down public transportation networks completely, there are significant distortions. As a rule of thumb, at least one third and up to two thirds of all connections in affected cities are canceled or severely delayed on strike days. After the official end of a strike, it usually takes some hours until service is back to normal.

Having described the context and setting of our study, we now go on to discuss how urban populations might be affected by public transit strikes.

⁶See U.S. Census Bureau, 2009-2013 American Community Survey 5-Year Estimates, Tables GCT0802, GCT0803 and GCT0804.

⁷Another restriction implicit in the German constitution is the so-called principle of *ultima ratio*. This principle represents the application of the general constitutional principle of proportionality (*Verhältnismäßigkeit*) in the field of labor law. According to this principle, a strike is only legal if it is necessary and the ultimate measure to solve an industrial conflict. Labor courts are empowered to assess the proportionality of industrial action and can, if necessary, sanction illegal strikes (Klaß *et al.*, 2008).

2.2. Public Transit Strikes and Car Traffic

Given the intensive use of public transportation in major German cities, we expect strikes by transit workers to have profound short-run effects on the mode of transport of commuters. Some might feel forced to use their private car or motorbike or a taxi on strike days. Others might switch to their bike or just walk. Again others might postpone their journey. Van Exel and Rietveld (2001) summarize the existing evidence as follows: public transit strikes induce most public transit users to switch to the car (either as driver or passenger) and as a result traffic density as well as road congestion increases. A similar conclusion is reached by Anderson (2014), who analyzes freeway traffic during a 35-day strike by transit workers in Los Angeles. His estimations reveal an increase in delays during peak periods by almost 50 percent due to increased car traffic.⁸ Finally, Adler and van Ommeren (2015) exploit transit strikes in Rotterdam and also find positive effects of transit shutdowns on congestion. Based on these findings we formulate our first testable prediction:

PREDICTION 1. Public transit strikes increase the number of cars on roads, especially during peak periods. Travel times increase due to rising traffic congestion.

2.3. Car Traffic and Accidents

The frequency and severity of road accidents depends on several traffic characteristics that may be affected by public transit strikes. Examples we have in mind include the number of cars in road systems, driving skills, driver behavior and speed. First, an often-used specification by transport economists suggests that the expected number of road accidents rises with the number of potential accidents which, in turn, is an increasing function of the number of cars in the system (Shefer and Rietveld, 1997). Edlin and Karaca-Mandic (2006) confirm this prediction by showing that traffic density increases accident costs substantially. Second, the expected number of road accidents is a function of the behavior and skills of drivers. In this regard, we would expect that public transit strikes reduce average driving skills since marginal drivers with less experience appear on road systems. This channel works to increase the frequency of road accidents. In addition, it is well understood that driving in high-density traffic can contribute to stress and therefore lead to behavioral patterns—e.g., tailgaiting, aggressive driving, breaking abruptly-that increase accident risk (OECD, 2007b). More accidents are likely to result in additional personal injuries (Shefer and Rietveld, 1997). However, the same logic does not necessarily apply to accidents involving severe injuries or fatalities: with an increase in congestion stemming from more cars in the system, average travel speed decreases, thus potentially causing

⁸Lo and Hall (2006) analyze the same strike using a simple before-after comparison, which has some methodological shortcomings as noted by Anderson (2014).

a reduction in the number of severe accidents. Evidence from the United States indeed suggests a substantial reduction in the number of fatal road accidents during morning peak hours, periods in which traffic density is the highest (Farmer and Williams, 2005). But there is also evidence, emerging from the United Kingdom, that the picture is more differentiated. In particular, congestion as a mitigator of crash severity is less likely to occur in urban conditions, but may still be a factor on higher speed roads and highways (Noland and Quddus, 2005). Our focus will be on accidents in urban conditions. Thus, it remains *a priori* unclear whether an increase in congestion stemming from public transit strikes affects the incidence of severe accidents, and if so in what direction. Against this background, our second testable prediction is:

PREDICTION 2. Public transit strikes increase the frequency of car accidents which, in turn, leads to a rise in accident-related injuries. The effect on accidents involving severe injuries or fatalities is a priori unclear.

2.4. Car Traffic and Air Pollution

Car traffic is associated with air pollution mainly due to engine exhaust. The chemical processes in fuel burning thus determine the expected effect of traffic on air pollution. Internal combustion engines powering the vast majority of cars in developed countries emit oxides of nitrogen, carbon monoxide, unburned or partially burned organic compounds and particulate matter with the amounts depending amongst other things on operating conditions (Heywood, 1988). In particular, it is well understood that congested stop-and-go traffic is associated with higher emissions than free-flow traffic. There are three reasons for this. First, the efficiency of internal combustion engines, which depends on revolutions per minute (rpm), is highest at medium speed (Davis and Diegel, 2007). Acceleration and deceleration episodes decrease the time operated in the optimal rpm range, which in turn increases emissions per minute driven. Second, congestion increases travel times, and so leads to a rise in fuel consumption and emissions per distance driven. Third, particulate matter emissions not only stem from fuel burning process, but also from break wear and tire wear on tarmac—both high in congested traffic. From an empirical viewpoint, several studies suggest that high traffic volumes and congestion are causes of ambient air pollution (see, e.g., Currie and Walker, 2011; Knittel et al., 2011). A pollutant which is not caused by car traffic, and therefore can be used for a placebo test, is sulfur dioxide. Indeed, sulfur dioxide emissions from cars are close to non-existent since modern gasoline no longer contains significant amounts of sulfur. From these arguments our third testable prediction arises:

PREDICTION 3. Public transit strikes increase road-traffic related air pollution. A pollutant expected to be unaffected is sulfur dioxide.

2.5. Air Pollution and Health

The exact pathophysiological effects of most air pollutants are not yet fully understood. However, a large body of research across many different disciplines suggests that exposure to air pollution can impair human health, even at pollution levels well below the limits set in developed countries (Beelen *et al.*, 2014). The identified effects range from respiratory symptoms and illness, impaired lung function, hospitalization for respiratory and cardiac disease to increases in mortality. The most harmful of the air pollutants stemming from car traffic is thought to be particulate matter. It is also widely accepted that infants and children are the subgroup of the population most susceptible to the effects of air pollution. This is mainly due to their ongoing respiratory development, smaller average lung size, and higher activity levels (Beatty and Shimshack, 2014). Furthermore, elderly people are at increased risk due to more frequent unfavorable health preconditions.

Much of what we know about pollution-related health problems is based on annual frequency data (see, e.g., Chay and Greenstone, 2003; Currie and Neidell, 2005; Currie et al., 2009). In contrast, our empirical analysis explores the daily, contemporaneous effect of public transit strikes on pollution-related health outcomes. This reduced-form is based on the idea that public transit strikes cause daily pollution shocks due to increased car traffic and congestion. Should we expect a short-term effect of air pollution on health? The existing evidence, while still relatively scarce, points towards an affirmative answer. Schlenker and Walker (2011) show that daily variation in ground level airport congestion due to network delays significantly increases both carbon monoxide emissions as well as hospital admissions for respiratory problems and heart disease. Their findings also suggest that infants and the elderly have a higher sensitivity to pollution fluctuations. In a similar vein, Atkinson et al. (1999) show that there is a positive association between daily emissions of particulate matter and daily visits to accident and emergency departments in London for respiratory complaints.⁹ Ransom and Pope (1992) exploit monthly variation in particulate matter emissions induced by the closure of a steel mill in Utah Valley, and find large effects on school absenteeism—a proxy for children's health. With this evidence in mind, we formulate our final testable prediction:

PREDICTION 4. Public transit strikes increase pollution-related health problems, especially among young children and the elderly.

⁹Relatedly, Schwartz and Dockery (1992) find that daily mortality in Philadelphia is positively associated with daily particulate matter pollution.

3. The Data

Our main sample spans the period from 2002 to 2011 and covers the five largest cities in Germany: Berlin, Hamburg, Munich, Cologne, and Frankfurt on the Main. We exploit six sources of data to analyze the extent to which the inhabitants of these cities are affected by public transit strikes.¹⁰

3.1. Strike Data

Our data on public transit strikes is self-collected and comes from newspaper archives, press releases of unions and official notifications of public transit operators. In order to ensure an accurate identification of strike activity, we employed a double-check procedure in the informationgathering process. In particular, we only coded a day as a strike day if congruent information from at least two independent sources indicated an episode of industrial action. During the sample period, from 2002 to 2011, unions calling strikes rarely resorted to lengthy campaigns of industrial action. Instead, the tactical use of one-day strikes was the norm.¹¹ We therefore only include public transit strikes of one day or less in our main sample, which leaves us with 71 incidences of strike activity across all cities.¹² The observed strikes either affect a city's local suburban train connections (S-Bahn) or its subway-tram-bus network.¹³ Figure 2 illustrates the distribution of strike activity across time and space. We observe 12 strikes in Berlin, 13 in Frankfurt-on-the-Main and Hamburg, 16 in Cologne, and 17 in Munich. At least one strike occurred in each year of the study period, and there were pronounced spikes in strike activity in 2007 and 2011. All strikes took place on weekdays, while weekends were unaffected. We observe strikes in all months of the year except in June and November. Finally, in our period of observation, unions rarely called strikes affecting all five cities simultaneously. Quite to the contrary, 20 strikes only affected a single city. In addition, 3 strikes affected two cities, 1 strike affected three cities, 3 strikes affected four cities, and on 6 occasions all five cities were simultaneously hit by a strike. Thus, we are able to exploit both time series and cross-sectional variation in our data. Table A1 of the Online Appendix provides detailed dates of all public

¹⁰The description of the data in the main body conveys core information only. In Online Appendix Table A2 we present detailed summary statistics.

¹¹In the data collection process, one specific reason for this tactic became apparent: strikes by German transit workers typically cause a great deal of initial disruption, but within a day or two of lengthier strikes many transit providers manage to implement effective emergency schedules which considerably dampen the impact of strikes.

¹²We also identified 17 public transit strikes with a duration of more than a day. The days affected by these lengthier strikes—amounting to a total of 74 city-day observations—are dropped from our main sample. In Section 5.7, we present regressions based on a sample including all strikes.

¹³In German cities, suburban train connections are run by Germany's rail operator *Deutsche Bahn*, while subway-tram-bus networks are operated by local transit providers. Workers respectively employed by *Deutsche Bahn* and local transit providers are represented by different unions, who usually do not call strikes simultaneously.

transit strikes in the sample period.

Our empirical analysis focuses on workweek days since this is when congestion occurs. Furthermore, there was no strike activity on weekends during the study period, as described above. Thus, we exclude weekends and public holidays¹⁴ from our data set.

3.2. Traffic Data

We obtained data on traffic volumes from the Federal Highway Research Institute (Bundesanstalt für Straßenwesen, BASt). Automated monitors operated by BASt collect hourly data on the number of passing vehicles on all freeways (Autobahnen) and non-freeway federal roads (Bundesstraßen) across Germany. The monitors are technically equipped to distinguish between car and truck traffic. Thus, we are able to execute a clean empirical test of the prediction that public transit strikes lead to an increase in car traffic. We include a total of 43 traffic monitors in our sample, all selected based on their locations on commuter routes into the cities of interest and their proximity to the respective city centers. In Figures A1-A5 in Online Appendix, we use the geocodes of the monitors to display their exact locations on city maps. As can be seen from the figures, 27 monitors are located on freeways, while 16 monitors are located on federal roads. The empirical analysis is based on hourly traffic data for the period January 1, 2002, to December 31, 2011. Due to maintenance work and upgrading, no data are available for Berlin from 2006 to 2010. Similarly, values are missing for Frankfurt-on-the-Main in 2004 and 2005. Figure A6 in Online Appendix shows how passenger vehicle flows change over the course of 24 hours for an average workweek day. There are two peak periods for car traffic. The first is between 6 a.m. and 10 a.m. in the morning when car traffic is nearly 85% higher as compared to the average hour. The second peak is between 3 p.m. and 7 p.m. in the afternoon. Based on these patterns, we define the morning peak (respectively, evening peak) to last from 6 a.m. to 10 a.m. (respectively, from 3 p.m. to 7 p.m.).

3.3. Congestion Data

We obtained data on traffic congestion from *TomTom*, a global supplier of location and navigation products and services. Since 2008, *TomTom* has been collecting anonymous GPS speed measurements from navigation users across cities around the globe.¹⁵ In a map-matching process, the GPS measurements are matched to digital city maps and assigned to road segments

¹⁴Public holidays also include the carnival days from Fat Thursday (*Weiberfastnacht*) to Shrove Tuesday (*Faschingsdienstag*), which can be regarded as *de facto* holidays.

¹⁵As of 2014, the GPS speed database contained 6 trillion measurements and grew by 6 billion measurements a day.

which vary in length between 2 meters and 2 km, depending on the complexity of the road situation. For our sample cities, speed measurements exist for road segments that add up to 3,637 km in Berlin, 2,263 km in Hamburg, 2,041 km in Munich, 1,988 km in Cologne, and 662 km in Frankfurt. When the map-matching process is complete, an aggregated geographic database (geobase) of measured road speeds is produced. These geobases are updated regularly for each map of each city to take into account the growing GPS speed database as well as changes in the road network (map). Each digital city map with attached speed information can be used to compute an average congestion index (CI) at daily frequencies. This index is defined as:

$$CI = \frac{T}{T_0}.$$

It compares actual travel times on all road segments in a city during the course of a day (T) to the free-flow travel times on these road segments (T_0). The difference is expressed as a percentage increase in travel time. Thus, a CI value of 1 implies that traffic was flowing freely throughout a day, while a value of 1.2 indicates that journeys took on average 20 percent longer than under non-congested conditions. In addition to the daily CI, we have access to daytime-specific CIs for the morning and the evening peak periods,¹⁶ as well as separate CIs for freeways and city streets. The CI data we obtained covers each city in our sample and spans the period from January 1, 2010 through to December 31, 2011. The average daily CI value is 1.3, which drops to 1.25 for highways and increases to 1.36 for city streets. As one would expect, the average CI values for the morning peak period, 1.47, and the evening peak period, 1.49, are higher than the average daily value.

3.4. Accident Data

Our information on accidents is based on register data which includes *all* vehicle crashes recorded by the German police. The police records are collected and made available by the statistical offices of the German states (*Statistische Landesämter*).¹⁷ Each police record includes a wide variety of information about the accident (such as time, date, location) together with a de-

¹⁶The congestion indices from the data provider are pre-defined variables that are aggregated at the city-day level. Peak morning congestion times are 8 a.m. to 9 a.m. on workweek days. Peak evening congestion times are 5 p.m. to 6 p.m. from Monday to Thursday for Hamburg, Munich, Cologne and Frankfurt and 4 p.m. to 5 p.m. for Berlin. On Fridays they are 3 p.m. to 4 p.m. for Berlin, Hamburg and Cologne and 5 p.m. to 6 p.m. for Munich and Frankfurt.

¹⁷The police does not forward records on minor accidents to the statistical offices, which are therefore not present in our database. Minor accidents are those in which (i) crashed vehicles remain in a roadworthy condition and (ii) all persons involved remain uninjured. In addition, the statistical offices do not provide access to information on vehicles crashes in which drivers were under influence of alcohol. Thus, alcohol-related accidents are not included in our database. Finally, our database does not include accidents in which the parties involved reached private agreements without involving the police.

scription of the number and types of injuries sustained in the accident. For the five cities in our sample, the police records available for the period 2002-2011 cover just over 354,400 vehicle crashes. We aggregate the police records to the city-day level while distinguishing between the morning and the evening peak hours. This procedure leaves us with a data set containing daily observations for the a.m. and p.m. peak period on (i) the number of vehicle crashes, (ii) the number of slightly injured persons, and (iii) the number of seriously or fatally injured persons.

3.5. Pollution Data

For the period 2002-2011, we obtained hourly data on atmospheric pollution from the Federal Environment Agency (*Umweltbundesamt, UBA*), which operates numerous air monitors across Germany. We include a total of 30 monitors in our sample, all selected based on their locations on streets within the five cities' boundaries.¹⁸ Figures A1-A5 of the Online Appendix show the locations of the monitors on city maps. We focus on two types of pollutants: inhalable coarse particles smaller than 10 micrometers in diameter (PM10) and nitrogen dioxide (NO2).¹⁹ In addition, we use sulfur dioxide (SO2) as a placebo pollutant in a falsification test. Figure A7 in Online Appendix shows how air pollution varies over the course of 24 hours for an average day of the workweek. For both PM10 and NO2, there are emission peaks during the morning and evening hours, respectively. We create pollution measures for the morning peak period (respectively, evening peak period) by taking the averages of all hourly readings between 6 a.m. and 10 a.m. (respectively, between 3 p.m. and 7 p.m.).

3.6. Hospitalizations: Diagnostic Data

We use data from the German hospitalization statistic for the years 2002-2010. The dataset provides information about *all* inpatients in *all* German hospitals. In particular, the following characteristics are collected for each patient: main diagnosis (3-digit ICD-10 code)²⁰, day of admission and discharge (day, month, year), place of residence (zip code, community), month and year of birth as well as gender. In order to examine pollution-related health problems, we focus on hospital admissions for diseases of the respiratory system (ICD-10 codes J00-J99) and abnormalities of breathing (ICD-10 code R06). For each type of diagnosis, we aggregate the number of hospitalizations by day of admission and patients' city of residence. Hence, we

¹⁸We exclude monitors that are situated around industrial areas, since these monitors capture air quality contaminant concentrations that relate to the industrial operators in the area.

¹⁹In an earlier version of this paper, we also examined carbon monoxide (CO) and found little evidence for a strike effect on this pollutant.

²⁰The ICD-10 classification ("International Statistical Classification of Diseases and Related Health Problems") categorizes diseases and other health problems recorded on many types of health and vital records.

obtain daily counts of hospitalizations, which we examine both for the entire population as well as for the population subgroups of those over 64 years of age and under 5 years of age.

3.7. Weather and Holiday Data

We obtained city-specific weather data at daily frequencies from the German Weather Agency (*Deutscher Wetterdienst*). In particular, we use daily measures of temperature, precipitation, wind speed, and a binary variable indicating snow cover to control for the direct effects of weather on the five outcomes of interest.²¹ To control for the direct effects of school holidays, we construct city-day dummy variables equal to unity when school holidays are in effect and zero otherwise. Our holiday data comes from the Standing Conference of the Ministers of Education and Cultural Affairs of the German states (*Kultusministerkonferenz*).

4. Empirical Strategy

Our identification strategy is based on a generalized difference-in-differences (DID) model which essentially compares outcomes in affected and non-affected cities before, during and after strike episodes. We now present our approach for regressions involving data at the monitor-hour level (car traffic). In this case, we estimate our basic specification as follows:

$$Y_{mchdwy} = \alpha + \beta (STRIKE_{cdwy}) + \gamma_h + \delta_d + \gamma_h \times \delta_d + \eta_w + \theta_y + \vartheta_m + \mu X_{cdwy} + \varepsilon_{mchdwy}.$$
 (1)

where Y_{mchdwy} is the number of cars passing monitor *m* in city *c* during hour *h* on day *d* in week *w* of year *y*. *STRIKE_{cdwy}* is a binary variable equal to unity when a strike is in effect and zero otherwise. We control for a full set of time fixed effects for each hour-of-day (γ_h), dayof-week (δ_d), week-of-year (η_w) and year (θ_y). Thus, we flexibly capture daytime and day-ofweek patterns, seasonal effects, and long-run time trends. The interactions between hour-of-day and day-of-week take into account that hourly traffic patterns might differ between days. By additionally including fixed effects for all monitors *m*, we account for time-constant differences between monitoring stations. The vector X_{cdwy} includes holiday and weather controls. In our preferred specification, we additionally allow for city-specific time fixed effects by including interactions of city indicators with hour-of-day, day-of-week, hour-of-day×day-of-week, weekof-year and year. Moreover, our preferred specification also controls for city-specific weather effects by interacting city indicators with all weather variables. When outcome variables are observed at the monitor-level with more than one station per city, we weight regressions by the

²¹Few missing observations for wind speed cause our number of observations to drop slightly when including controls.

inverse of the number of observations in each city. This weighting procedure ensures that each city is given the same weight in the regressions. For regressions involving data aggregated to the monitor-day level (air pollution), we drop hour-of-day fixed effects and their interactions. For data aggregated to the city-day level (congestion, accidents, health), we additionally replace monitor fixed effects with city fixed effects.

In our setting, standard errors might be biased due to serial correlation. We therefore follow Bertrand *et al.* (2004) in clustering standard errors at the city level, the highest aggregation level where correlation may occur. In order to account for the small number of clusters, the Wald test uses a conservative T(G-1) distribution to compute p-values, with G being the number of clusters. Since the ad-hoc corrections for few clusters might still understate the true size of the standard errors, we also check whether our results hold using wild cluster bootstrap t-procedures (Cameron *et al.*, 2008). To do so, we create pseudo-samples applying cluster-specific Rademacher weights (+1 and -1 with equal probabilities) to the residuals of the original regression under the null hypothesis of no strike effect. We then estimate the strike effect on the pseudo-samples holding the vector of controls constant. Thus, we receive a distribution of t-values, which is finally used for statistical inference. In the results section, we will focus on models using clustered standard errors to draw statistical inference. However, virtually all findings are confirmed if we instead use wild cluster bootstrap t-procedures.

We assume that conditional on the covariates, the location and timing of strike activity is orthogonal to traffic volumes, travel times, accident risk, pollution emissions, and health. A potential threat to identification arises if public transit strikes are planned to cause maximum disruption. If this is the case, one might expect the timing of strikes to coincide with hours of the day and/or days of the week during which traffic density is the highest. Note, however, that we control for this type of confounding variation by including hour-of-day and day-of-week fixed effects as well as the interaction between them. Union leaders may also choose to initiate strikes at location-time combinations where they are likely to cause maximum disruption. In our most extensive specification, we account for this possibility by including a full set of city-specific time fixed effects in addition to the monitor or city fixed effects. There are other occasions where the impact of strikes is conceivably high: at the beginning of holidays or during periods of bad weather. Again, these candidate confounders are controlled for. In addition to suitable conditioning, we conduct a number of sensitivity checks to support our design and identifying assumption. In particular, we examine whether the estimated effects of interest are robust to the inclusion of additional city-specific time-varying covariates (e.g., mass events). Moreover, we provide evidence from regressions involving both placebo strikes as well as placebo outcomes.

5. Results

5.1. Car Traffic

Table 1 reports the results for passenger vehicle flows. The first panel presents regression estimates involving only morning peak period data for freeways. The morning peak is defined to last from 6 a.m. to 10 a.m. Column (1) estimates Equation (1) conditioning only on monitor fixed effects and the full set of time fixed effects. In the morning peak hours of strike-free days, the average hourly traffic flow on freeways amounts to 5,239 passenger vehicles per monitor. During a strike, vehicle flows in the morning increase by 161 cars per hour and monitor, an effect significant at the 1% level. Column (2) shows the result to be robust to including controls for local weather conditions and school holidays. In Column (3), we interact the full set of time fixed effects with city indicators. Controlling for city-specific time effects in this way leaves the estimated strike effect largely unchanged. In Column (4) we additionally interact the full set of weather controls with city indicators. The strike coefficient remains virtually unaffected and highly significant. The estimate from our preferred specification in Column (4) suggests that public transit strikes lead to an increase in car traffic during the a.m. peak period by 2.5%. The second panel repeats the exercise for federal roads. The estimate from our preferred specification suggests a 4.3% increase in car traffic on federal roads during the a.m. peak of a strike day (Column (4)). The last two panels of Table 1 shows the strike effects in the evening peak hours from 3 p.m. to 7 p.m. Throughout all specifications, the strike effect turns out positive and significant for freeways. Moreover, in our preferred specification, the strike effect also gains statistical significance for federal roads. We observe that the estimates are somewhat smaller in size during the p.m. peak period than during the a.m. peak period, suggesting an increase in traffic flows by slightly less than 2% both on freeways and federal roads.

Our data also allows us to provide a picture of strike impact over the course of a day. In Figure 3, we plot the results of a regression interacting our strike indicator with *all* hours of the day. For periods outside the morning and evening peak, strikes in public transportation leave traffic volumes virtually unaffected. For freeway traffic (Panel (a)), significant hourly strike effects arise between 5 a.m. and 10 a.m. as well as between 1 p.m. and 7 p.m. The most pronounced effect arises in the morning between 6 a.m. and 7 a.m., when traffic volumes increase by 7.7%.²² Compared to the a.m. peak effect of strikes, the p.m. peak effect is smaller but spreads out over a longer period. This might occur because commuters usually have more flexibility in decisions over departure time in the evening than in the morning commute. For traffic on federal roads (Panel (b)), the most pronounced strike effect also arises between 6 a.m.

²²The effect size is 344 cars and the average number of cars during that hour is 4,477.

and 7 a.m., when traffic volumes increase by 9.4%.²³ Moreover, the a.m. peak effects are again more pronounced than the p.m. peak effects.

Despite our flexible estimation approach, it is important to acknowledge the possibility that traffic is unusually high on strike days for reasons other than strikes being in effect. We now conduct a falsification test in order to rule out such confounding bias in our design. Recall that most strikes in our sample did not affect all five cities simultaneously. This allows us to geographically shift $STRIKE_{cdwy}$ from cities affected by strikes to non-affected cities. If our design is valid, then there should be no significant effects on car traffic in these non-affected cities.²⁴

For graphical inspection, we first compute the residuals of a regression of the number of cars per hour on the most extensive set of control variables under the null hypothesis of no strike effect. We then plot the residuals of the number of cars per hour against hours, where 6 a.m. of a strike day is normalized to zero.²⁵ Thus, data points represent hourly averages of unexplained variations in vehicle flows. Based on these data points, we apply local polynomial smoothing techniques.²⁶ As is evident from the first panel of Figure 4, there is no jump in car traffic on freeways in non-affected cities when strikes begin elsewhere. Indeed, the unexplained variations in vehicle flows run absolutely smoothly across the placebo strike threshold. For affected cities, by contrast, there is a significant upward jump in car traffic when strikes begin, as can be seen in the second panel. Apart from this jump at the strike threshold, unexplained variations in vehicle flows are remarkably flat within a period of three weeks before and three after a strike episode. The last two panels of Figure 4 repeat the exercise for federal roads and return qualitatively identical results. Table 2 presents the placebo analog of Table 1. Across all specifications, and for both morning and evening peak hours, the placebo effect of public transit strikes on vehicle flows on both freeways and federal roads is statistically insignificant²⁷ and small in magnitude, fluctuating around zero.

5.2. Travel Times

Table 3 presents regressions estimating the effect of transit strikes on travel times. The dependent variables are congestion indices (CIs) based on *TomTom*'s GPS speed database. The first panel reports results using the CI for the morning peak period. The estimate from our preferred

²³The effect size is 140 cars and the average number of cars in during that hour is 1,484.

²⁴Observations from struck cities are excluded from the placebo sample in order to exclude bias on the placebo control dates.

²⁵For presentational reasons, we exclude data points left and right of the discontinuity that were also strike days.

²⁶We follow Lee and Lemieux (2010) and use a rectangular kernel for the smoothing function with first order polynomials and a bandwidth of 48 hours.

²⁷One exception is the estimate in the minimum specification for federal roads during the morning peak.

specification indicates that the CI for the morning peak period increases by 0.123, which implies that average morning travel times increase by 8.4%. As can be seen across Columns (1) to (4), the sign, magnitude and significance of the coefficient on our strike indicator is very robust across the four specifications. The second panel presents analogous estimates for the evening peak period, which are smaller in magnitude than the effects during the morning hours. Evaluated against the average evening CI of 1.49, the results suggest that average travel times in the evening increase between 3.7% and 4.3%, although the coefficient reported in Column (4) loses statistical significance. In the third panel, results for the average peak period hour are depicted. The preferred specification in Column (4) implies a significant increase of travel times by 6.3%. The fourth panel reports results of regressions using the CI averaged over the day as the dependent variable. The results suggest that strikes increase average travel times between 3.8% to 4.3% over the course of a day. All estimates turn out to be statistically significant. In the last two panels, we use daily CIs for inner-city streets and highways, respectively. While the effects for city streets are more precisely estimated than for highways, the point estimates are almost identical and, depending on the specification, suggest increases in average travel times between 3.4% and 4.4%. Thus, strike-induced congestion spreads over all types of streets within cities and is not exclusive to freeways or inner-city streets.

5.3. Total Car Hours Operated

In what follows, we will further investigate the effects of public transit strikes on accident risk and pollution emissions. Both outcomes are likely to depend on total car hours operated, which in turn are determined by the number of vehicles on roads and average travel time:

 $[\text{total car hours operated}] = [\# \text{ cars on roads}] \times [\emptyset \text{ travel time in hours}].$

Our results so far suggest that strikes by transit workers affect both terms on the right hand side of this equation, with the effects being strongest during the morning peak period. Indeed, during that period, strikes increase the number of passenger vehicles on roads by 2.5% (freeways) to 4.3% (federal roads) and raise travel times by 8.4%. Both effects combine according to:

$$\% \Delta[\text{total car hours operated}] = \left[\left(1 + \frac{\% \Delta[\# \text{ cars on roads}]}{100} \right) \left(1 + \frac{\% \Delta[\emptyset \text{ travel time in hours}]}{100} \right) - 1 \right] \times 100.$$

During the a.m. peak period, public transit strikes therefore lead to a 11% to 13% increase in total car hours operated. This is the benchmark against which we will evaluate subsequent results on accident risk and pollution emissions.

5.4. Vehicle Crashes and Accident-Related Injuries

Table 4 reports the effects of public transit strikes on vehicle crashes and accident-related injuries. The first panel uses the number of vehicle crashes during the a.m. peak period as the dependent variable. In the morning peak hours of strike-free days, there are on average 4.28 vehicle crashes per city. During a strike, the number of vehicle crashes in the morning hours increases by 0.607, or 14.2% of the strike-free level (Column (4)). This increase is statistically significant and remains very stable regardless of the specification. The second panel reports the results for the number of persons sustaining slight injuries in vehicle crashes. Focusing on our preferred specification (Column (4)), we find that strikes significantly increase the number of slightly injured persons by 0.790. Compared to the 3.94 personal injuries we observe during the morning hours of strike-fee days, this corresponds to a 20.1% increase. The fact that the increase in personal injuries exceeds the increase in the number of vehicles crashes suggests that cars are occupied by more passengers on strike days. This is consistent with evidence suggesting that strikes induce public transit users to switch to the car either as driver or passenger (Van Exel and Rietveld, 2001). Finally, there is no significant effect on the number of seriously or fatally injured persons, as is evident from the results reported in the third panel. In the last three panels of Table 4, we repeat the exercise using accident data for the p.m. peak period. All evening estimates on vehicles crashes and accident-related injuries are statistically insignificant.

5.5. Air Pollution

Table 5 contains two central results on air pollution. The estimates in the first panel indicate that public transit strikes have a statistically significant and positive effect on morning peak emissions from particulate matter, a major traffic-related pollutant. In particular, the results in Columns (1) to (4) imply that particle pollution increases by 13.3% to 14.8% during the a.m. peak hours of a strike day. The results in the second panel suggest that public transit strikes also have positive effects on morning peak emissions of nitrogen dioxide. For example, our preferred specification (Column (4)) yields a statistically significant increase of NO2 by $3.31 \ \mu g/m^3$, or 4.3% of the strike-free level. In the last two panels of Table 5, we repeat the exercise using pollution data for the p.m. peak period. All evening estimates on air pollution are statistically insignificant.

A potential threat to identification is that air pollution on strike days might be higher than usual for reasons other than strikes being in effect. Although we control for an extensive set of time fixed effects and local weather conditions, there may be unobserved time-varying factors that are correlated with strikes and at the same time determine the occurrence and durability of pollutants in ambient air. To empirically analyze the relevance of these concerns, we now conduct another falsification test. In particular, we investigate the effect of public transit strikes on SO2. As mentioned above, sulfur dioxide is no longer a major tailpipe pollutant. However, it nevertheless depends on environmental conditions like many other pollutants. Table 6 reports the results. Across all specifications, and for both morning and evening peak hours, the effect of public transit strikes on SO2 is statistically insignificant, which corroborates the validity of our empirical design.

5.6. *Hospitalizations*

Table 7 reports the results for pollution-related health problems. The first panel presents regression estimates involving data on hospitalizations for diseases of the respiratory system. On an average strike-free day, we observe 61 hospital admissions for respiratory illnesses per city, roughly 8 of which occur among children under 5 years of age. On a strike day, the number of children diagnosed with respiratory illnesses increases by 0.879, or 11% of the strike-free level (Column (2)). The estimate is statistically significant. At the same time, there is no evidence for an increase in respiratory illnesses in the total population or in the subgroup of the elderly (Columns (1) and (3)). The second panel uses hospitalizations for abnormalities of breathing as the dependent variable. On an average strike day, the total number of patients admitted to hospitals due to breathing problems increases by 13% (Column (1)), an estimate significant at the 5% level. As before, the effect appears to be driven by the subgroup of young children, for whom we find a precisely estimated 34% increase in hospital admissions for abnormalities of breathing (Column (2)). The strike dummy variable for the elderly patient subgroup has a positive but not statistically significant coefficient. We also examined hospital admissions for diseases of the circulatory system (ICD-10 codes I00-I99). We found no evidence for a strike effect on circulatory illnesses.

5.7. Robustness

Mass Events. Our estimates in the previous section would be biased if there were omitted variables that are correlated with the occurrence of strikes and the outcomes of interest. For example, suppose that strikes by transit workers tend to coincide with mass events (e.g., trade fairs, sporting events, festivals). If mass events result in an increase (respectively, decrease) in traffic volumes, then omitting controls for such events results in an upward (respectively, downward) biased estimate of the true effect of public transit strikes. To mitigate this omitted variable bias, we now extensively control for mass events at the city-day level. In particular, we add the binary variable (*MassEvent_{cdwy}*) to Equation (1), which equals unity for events such as the Beer Festival (*Oktoberfest*) and Security Conference in Munich, the Harbor Festival (*Hafenfest*) in

Hamburg, the Museum Embankment Festival (*Museumsuferfest*) in Frankfurt on the Main, the Christopher Street Day Parade in Cologne, or the Carnival of Cultures (*Karneval der Kulturen*), the Fan Park during the 2006 Soccer World Championship in Berlin and a number of trade fairs.²⁸ The results reported in Table 8 show that the mass event coefficients turn out to be small throughout all specifications and mostly insignificant. More importantly, the coefficients on our strike dummy variable remain virtually unchanged compared to the benchmark estimates in Table 1.

Multi-Day Strikes. We have so far exploited 71 one-day strikes in public transportation over the period from 2002 to 2011. During that period, there were also 17 strikes with a duration of more than one day across the five cities. We now add all workweek days affected by these multi-day strikes—amounting to a total of 74 city-day observations—to our sample. Then, we re-estimate Equation (1) using both a one-day strike dummy and a multi-day strike dummy as independent variables. Table 9 presents the results of this extended specification for passenger vehicle flows.²⁹ The estimates suggest that the effect of multi-day strikes on car traffic is generally smaller than the effect of one-day strikes. One possible explanation for this result is one that we already mentioned: strikes by transit workers in Germany cause a great deal of initial disruption, but within a day or two of lengthier strikes transit provider typically manage to implement effective emergency schedules which dampen the impact of strikes.

Standard Errors. Since reliable inference is a concern when there are few clusters, we checked whether our results also hold using wild cluster bootstrap t-procedures instead of clustering standard errors. As mentioned above, all our findings were very robust to using the standard 2-point wild cluster bootstrap suggested by Cameron *et al.* (2008). However, Webb (2014) argues that this procedure may be noisy with a small number of clusters because the estimated p-values are intervals rather than point estimates. In order to receive more precise p-values, he suggests expanding the standard 2-point wild cluster bootstrap to a multi-point wild cluster bootstrap. We followed this suggestion and substituted the Rademacher weights (+1 and -1 with equal probabilities) by randomly drawing the weights from a normal distribution with a mean of zero and a standard deviation of one. The p-values obtained from this alternative bootstrap procedure suggest that the estimated strike effects gain rather than lose statistical significance.

²⁸The extended model controls for a total of 55 mass events across the five cities, attracting crowds of more than 150,000 people per day on average. The days affected by these events amount to a total of 1,091 city-day observations.

²⁹The resulting sample for freeways includes 64 one-day strikes and 12 multi-day strikes, which cover 41 cityday observations. In the sample for federal roads, we observe 45 one-day strikes and 10 multi-day strikes, which cover 37 city-day observations.

Measurement Error. Our strike indicator is based on self-collected data and might therefore be prone to measurement error. Indeed, we cannot entirely rule out that we (i) missed days that were affected by strikes or (ii) erroneously coded a day as a strike day even though no strike took place. Note, however, that both types of measurement error would result in a downward bias in the estimated effects of public transit strikes. If we missed days that were affected by strikes and hence erroneously coded them as non-strike days, then car traffic on non-strike days would be higher, which in turn reduces the estimated effect of strikes. If we erroneously coded a day as a strike day, car traffic on strike days would be lower, which again reduces the estimated effect of strikes.

6. Discussion and Conclusions

How large are the costs of transit strikes to non-involved third parties? The lion's share of thirdparty costs stems from the increase in travel time due to congestion. From the 2003 wave of the German Socio-Economic Panel (SOEP)³⁰, we obtain information on commuter incidence, modes of transport, and travel times. In the five cities of our sample, 47% of the working population commute to their work place using a car, while 43% rely on public transit. Combining this information with local employment data,³¹ the average number of car commuters per city amounts to 486,000, while there are on average 445,000 commuters using mass transit. According to the SOEP, average one-way travel-to-work time is 27 minutes for car commuters and 37 minutes for commuters relying on public transit. The estimates in Table 3 imply that travel times for car commuters increase by 6.3% the during the peak periods. We assume that mass-transit commuters experience the same percentage increase in travel times as car commuters, irrespective of whether they switch to the car or continue to use public transport on strike days. In the average city, a single one-day strike therefore implies an increase in aggregate travel time by roughly 62,000 hours, or 1,550 full-time equivalent work weeks. Valuing time at average GDP per hour worked, $\in 52^{32}$ we estimate congestion costs of $\in 3.2$ million per strike or $\in 228.9$ million for all 71 strikes in our main sample.

If these costs are not internalized in the collective bargaining process, the level of strike activity resulting from failed negotiations will be inefficiently high. In this regard, it might be interesting to set the third-party congestion costs in relation to the costs of struck employers.

³⁰Socio-Economic Panel (SOEP), Data from 1984-2012, DOI: 10.5684/soep.v29.1. We use the SOEPremote version to identify the cities of our sample.

³¹Average number of employed individuals per city is 1,043,000. See Statistical Offices of Federal State and States (2011), working population, http://aketr.de/tl_files/aketr/DATA/Tabellen/KR_ET.pdf, as of 03/26/2014.

³²See Statistical Offices of Federal State and States (2011), http://www.vgrdl.de/Arbeitskreis_VGR/tbls/R2B1.zip and http://aketr.de/tl_files/aketr/DATA/Tabellen/KR_AV.pdf, as of 03/26/2014.

For transit providers, the withdrawal of striking workers means a partial shutdown of services, and with it a loss of revenues from ticket sales. In the average city, transit providers generate revenues from ticket sales of \in 445.8 million annually. Assuming that struck transit providers are unable to raise any revenue from their users, this corresponds to a revenue loss of \in 1.2 million per strike day, or roughly one-third of the daily congestion costs to non-involved third parties.

Our most interesting and novel finding is that strikes in public transportation not only cause congestion costs, but also pose a non-negligible threat for public safety and public health. We have shown that public transit strikes cause daily pollution shocks accompanied by an increase in pollution-related health problems. For children under 5 years of age, hospital admissions for respiratory diseases and abnormalities of breathing increase by 11% and 34%, respectively. With 71 transit strikes in our sample, 68 more young children had to be admitted to hospitals than would have been if there had been no strikes. Moreover, our estimates suggest that transit strikes increase the risk of being injured in a motor vehicle crash by 20%. According to the International Labour Organization (ILO), governments can ban strikes in "essential services", defined as a service whose stoppage poses a clear and imminent threat to the life, personal safety or health of the whole or part of the population. Public transportation does not fall under the ILO's definition of an essential service. Taken at face value, our results seem to provide strong evidence in support of the opposite position: that mass transit—just as the police or firefighters—is critical to public safety and health on a day-to-day basis.

It is important to keep a few caveats in mind. Our analysis leaves open the question of whether laws banning public transportation strikes are welfare-enhancing. Tracing the total welfare consequences of strikes is complex. Our analysis shows that strike-induced disruptions of mass transit services have adverse effects on urban populations in the short-run. However, it misses any longer-run impacts of public transit strikes. For example, it stands to reason that these strikes may provide offsetting long-term benefits for urban populations if they result in agreements that improve organizational performance in urban mass transit. Further research is therefore warranted to develop a comprehensive approach for establishing a measure of the welfare effects of strikes in public transportation.

Another issue is external validity. It seems reasonable to assume that the size of the impact of transit strikes on the studied outcomes depends on several mediating factors. The following examples spring to mind: the capacity of highways and roads to absorb additional drivers; the average age of cars on roads; environmental laws regulating car emissions; posted speed limits; or prominent weather features that affect the accumulation of pollution. These mediating factors are likely to vary across jurisdictions. Thus, the estimated strike effects in German cities might be different from similar strikes in, say, US cities, which we have cited as a points of comparison. In order to gauge the external applicability of our results, future research should therefore attempt to document how the impact of public transit strikes varies along mediating factors.

References

- ADLER, M. W. and VAN OMMEREN, J. N. (2015). Does Public Transit Reduce Car Travel Externalities? *Tinbergen Institute Discussion Paper 2015-011/VIII*.
- ANDERSON, M. L. (2014). Subways, Strikes, and Slowdowns: The Impacts of Public Transit on Traffic Congestion. *American Economic Review*, **104** (9).
- ATKINSON, R., ANDERSON, H., STRACHAN, D., BLAND, J., BREMMER, S. and PONCE DE LEON, A. (1999). Short-term associations between outdoor air pollution and visits to accident and emergency departments in london for respiratory complaints. *European Respiratory Journal*, **13** (2), 257–265.
- BAKER, M. (2013). Industrial Actions in Schools: Strikes and Student Achievement. *Canadian Journal of Economics*, **46** (3), 1014–1036.
- BEATTY, T. K. and SHIMSHACK, J. P. (2014). Air Pollution and Children's Respiratory Health: A Cohort Analysis. *Journal of Environmental Economics and Management*, **67** (1), 39–57.
- BEELEN, R., RAASCHOU-NIELSEN, O., STAFOGGIA, M., ANDERSEN, Z. J., WEINMAYR, G., HOFFMANN, B., WOLF, K., SAMOLI, E., FISCHER, P., NIEUWENHUIJSEN, M. *et al.* (2014). Effects of Long-Term Exposure to Air Pollution on Natural-Cause Mortality: an Analysis of 22 European Cohorts within the Multicentre ESCAPE Project. *The Lancet*, 383 (9919), 785–795.
- BELOT, M. and WEBBINK, D. (2010). Do Teacher Strikes Harm Educational Attainment of Students? *Labour*, 24 (4), 391–406.
- BERTRAND, M., DUFLO, E. and MULLAINATHAN, S. (2004). How Much Should We Trust Differences-In-Differences Estimates? *The Quarterly Journal of Economics*, **119** (1), 249– 275.
- CAMERON, A. C., GELBACH, J. B. and MILLER, D. L. (2008). Bootstrap-Based Improvements for Inference with Clustered Errors. *Review of Economics and Statistics*, **90** (3), 414– 427.

- CHAY, K. Y. and GREENSTONE, M. (2003). The Impact of Air Pollution on Infant Mortality: Evidence from Geographic Variation in Pollution Shocks Induced by a Recession. *The Quarterly Journal of Economics*, **118** (3), 1121–1167.
- CHEN, Y. and WHALLEY, A. (2012). Green Infrastructure: The Effects of Urban Rail Transit on Air Quality. *American Economic Journal: Economic Policy*, **4** (1), 58–97.
- CUNNINGHAM, S. A., MITCHELL, K., VENKAT NARAYAN, K. and YUSUF, S. (2008). Doctors' Strikes and Mortality: A Review. *Social Science & Medicine*, **67** (11), 1784–1788.
- CURRIE, J., HANUSHEK, E. A., KAHN, E. M., NEIDELL, M. and RIVKIN, S. G. (2009). Does pollution increase school absences? *The Review of Economics and Statistics*, **91** (4), 682–694.
- and NEIDELL, M. (2005). Air Pollution and Infant Health: What Can We Learn from California's Recent Experience? *The Quarterly Journal of Economics*, **120** (3), 1003–1030.
- and WALKER, R. (2011). Traffic congestion and infant health: Evidence from e-zpass. *American Economic Journal: Applied Economics*, **3** (1), 65–90.
- DAVIS, S. C. and DIEGEL, S. W. (2007). *Transportation Energy Data Book*, vol. 26. Oak Ridge: ORNL-6978, Prepared for the US Department of Energy.
- DURANTON, G. and TURNER, M. A. (2011). The Fundamental Law of Road Congestion: Evidence from US Cities. *American Economic Review*, **101** (6), 2616–52.
- EDLIN, A. S. and KARACA-MANDIC, P. (2006). The Accident Externality from Driving. *Journal of Political Economy*, **114** (5), 931–955.
- FARMER, C. M. and WILLIAMS, A. F. (2005). Temporal Factors in Motor Vehicle Crash Deaths. *Injury Prevention*, **11** (1), 18–23.
- GERNIGON, B., ODERO, A. and GUIDO, H. (1998). ILO Principles Concerning the Right to Strike. *International Labour Review*, **137** (4), 441–481.
- GROSS, J. A. (1999). A Human Rights Perspective on United Labor Relations Law: A Violation of the Right of Freedom of Association. *Employee Rights and Employment Polciy Journal*, 3 (1), 65–103.
- GRUBER, J. and KLEINER, S. A. (2012). Do Strikes Kill? Evidence from New York State. *American Economic Journal: Economic Policy*, **4** (1), 127–157.

- HEYWOOD, J. B. (1988). *Internal Combustion Engine Fundamentals*, vol. 930. New York: McGraw-Hill.
- ILO (2006). Freedom of Association: Digest of Decisions and Principles of the Freedom of Association Committe of the Governing Body of the ILO. Fifth (revised) edition. Geneva.
- ILO (2011). *362nd Report of the Committee on Freedom of Association*. International Labor Office, Governing Body, 312th Session, Geneva.
- KAUFMANN, B. E. (1992). Research on Strike Models and Outcomes in the 1980s: Accomplishments and Shortcomings. In D. Lewin, O. S. Mitchell and P. D. Sherer (eds.), *Research Frontiers in Industrial Relations and Human Resources*, Madison: Industrial Relations Research Association, pp. 77–129.
- KLASS, F., RÖLZ, H., RABE, S. and REITEMEYER, S. (2008). Germany. In A. Stewart and M. Bell (eds.), *The Right to Strike: A Comparative Perspective*, Liverpool: The Institute of Employment Rights, pp. 53–64.
- KNITTEL, C. R., MILLER, D. L. and SANDERS, N. J. (2011). Caution, Drivers! Children Present: Traffic, Pollution, and Infant Health. *NBER Working Paper No. 17222*.
- KRUEGER, A. B. and MAS, A. (2004). Strikes, scabs, and tread separations: Labor strife and the production of defective bridgestone/firestone tires. *Journal of Political Economy*, **112** (2), 253–289.
- LALIVE, R., LUECHINGER, S. and SCHMUTZLER, A. (2013). Does Supporting Passenger Railways Reduce Road Traffic Externalities? *CEPR Discussion Paper No. 9335*.
- LO, S.-C. and HALL, R. W. (2006). Effects of the Los Angeles Transit Strike on Highway Congestion. *Transportation Research Part A: Policy and Practice*, **40** (10), 903–917.
- MAS, A. (2008). Labour Unrest and the Quality of Production: Evidence from the Construction Equipment Resale Market. *The Review of Economic Studies*, **75** (1), 229–258.
- NELSON, P., BAGLINO, A., HARRINGTON, W., SAFIROVA, E. and LIPMAN, A. (2007). Transit in Washington, DC: Current Benefits and Optimal Level of Provision. *Journal of Urban Economics*, **62** (2), 231–251.
- NOLAND, R. B. and QUDDUS, M. A. (2005). Congestion and Safety: A Spatial Analysis of London. *Transportation Research Part A: Policy and Practice*, **39** (7), 737–754.

- NYS DEPARTMENT OF STATE (2009). *Local Governemnt Handbook: 6th Edition*. New York State Department of State.
- OECD (2007a). *Infrastructure to 2030: Mapping Policy for Electricity, Water and Transport.* Paris: Organisation for Economic Co-operation and Development.
- OECD (2007b). *Managing Urban Traffic Congestion*. Paris: Organisation for Economic Cooperation and Development.
- PANTELL, R. H. and IRWIN, C. E. (1979). Appendectomies During Physicians' Boycott: Analysis of Surgical Care. *The Journal of the American Medical Association*, **242** (15), 1627– 1630.
- RANSOM, M. R. and POPE, C. A. (1992). Elementary school absences and pm10 pollution in utah valley. *Environmental Research*, **58** (1), 204–219.
- SCHLENKER, W. and WALKER, W. R. (2011). Airports, Air Pollution, and Contemporaneous Health. *NBER Working Paper No. 17684*.
- SCHWARTZ, J. and DOCKERY, D. W. (1992). Increased Mortality in Philadelphia Associated with Daily Air Pollution Concentrations. *American Review of Respiratory Disease*, **145** (3), 600–604.
- SHEFER, D. and RIETVELD, P. (1997). Congestion and Safety on Highways: Towards an Analytical Model. *Urban Studies*, **34** (4), 679–692.
- SWEARENGEN, K. M. (2010). Tailoring the taylor law: Restoring a balance of power to bargaining. *Columbia Journal of Law and Social Problems*, **44**, 513–551.
- VAN EXEL, N. and RIETVELD, P. (2001). Public transport strikes and traveller behaviour. *Transport Policy*, **8** (4), 237–246.
- WEBB, M. D. (2014). Reworking Wild Bootstrap Based Inference for Clustered Errors. Queen's Economics Department Working Paper No. 1315.
- WINGERTER, C. (2014). Berufspendler: Infrastruktur wichtiger als Benzinpreis. STATmagazin:Arbeitsmarkt, 5, 1–4.



FIGURE 2: Distribution of Strikes Across Time and Space

29



FIGURE 3: The Hourly Effect of Strikes on Car Traffic

NOTES: The grey bars show the hourly mean of the number of cars passing monitoring stations from Monday to Friday. The black whiskers indicate the 95% confidence interval of the hourly strike effects which are added to the hourly mean numbers of cars. The strike effects are estimated in a regression controlling for monitor fixed effects and the full set of time fixed effects. Additional controls are the amount of precipitation (and its square), days since last rainfall, atmospheric temperature (and its square), wind speed (and its square), and a snow cover dummy, and interactions of the full set of time fixed effects and the weather variables with city indicators. Standard errors are clustered at the city level.



Dependent Variabl	le: Hourly Passenger	Vehicle Flows per Monit	or	
	(1)	(2)	(3)	(4)
		1. Freeways –	Morning Peak	
Strike	160.7***	136.8***	128.3***	131.6***
[5,239]	(26.81)	(10.81)	(13.79)	(12.07)
$\frac{N}{R^2}$	213,160	212,896	212,896	212,896
	0.899	0.903	0.921	0.922
		2. Federal Roads	s – Morning Peak	
Strike	62.93**	57.38**	72.40***	77.72***
[1,790]	(12.82)	(15.33)	(7.99)	(9.49)
$\frac{N}{R^2}$	102,704	102,540	102,540	102,540
	0.921	0.924	0.961	0.962
		3. Freeways –	Evening Peak	
Strike	125.4**	103.3***	88.35**	91.50***
[5,785]	(28.61)	(21.24)	(19.72)	(17.89)
$\frac{N}{R^2}$	213,160	212,896	212,896	212,896
	0.937	0.939	0.950	0.950
		4. Federal Road	s – Evening Peak	
Strike	21.44	13.71	26.29	37.88**
[2,121]	(17.38)	(18.12)	(14.29)	(10.18)
$\frac{N}{R^2}$	102,704	102,540	102,540	102,540
	0.960	0.962	0.972	0.973
Monitor FE	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes
Controls City×Time City×Weather		Yes	Yes Yes	Yes Yes Yes

TABLE 1: The Effect of Strikes on Car Traffic

NOTES: Number of one-day strikes used in estimation sample: 64 for freeways, 45 for federal roads. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include monitor fixed effects. Time FE include hour-of-day, day-of-week, hour-of-day×day-of-week, week-of-year, and year. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City×Time are interactions of city indicators with all Time FE. City×Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variable	e: Hourly Passenger	Vehicle Flows per Mon	itor	
	(1)	(2)	(3)	(4)
		1. Freeways –	Morning Peak	
Placebo Strike	-17.52	-28.42	-22.12	-16.21
[5,239]	(24.47)	(41.47)	(36.84)	(35.33)
N	211,836	211,572	211,572	211,572
R^2	0.899	0.903	0.921	0.922
		2. Federal Roads	– Morning Peak	
Placebo Strike	14.70*	7.171	-7.729	-6.846
[1,790]	(5.563)	(5.947)	(12.91)	(10.66)
Ν	102,124	101,960	101,960	101,960
R^2	0.921	0.924	0.961	0.962
		3. Freeways –	Evening Peak	
Placebo Strike	-17.07	-15.83	-11.85	-4.038
[5,785]	(10.72)	(18.24)	(22.23)	(20.04)
Ν	211,836	211,572	211,572	211,572
R^2	0.937	0.939	0.950	0.950
		4. Federal Roads	s – Evening Peak	
Placebo Strike	-8.00	-10.86	-21.52	-17.56
[2,121]	(5.867)	(11.28)	(18.75)	(16.15)
Ν	102,124	101,960	101,960	101,960
R^2	0.960	0.962	0.972	0.973
Monitor FE	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes
Controls		Yes	Yes	Yes
City× 11me			res	res Voc
City × weather				105

TABLE 2: The Effect of Placebo Strikes on Car Traffic

NOTES: Number of one-day placebo strikes used in estimation sample: 74 for freeways, 61 for federal roads. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include monitor fixed effects. Time FE include hour-of-day, day-of-week, hour-of-day×day-of-week, week-of-year, and year. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City×Time are interactions of city indicators with all Time FE. City×Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variable	e: Actual Travel Tim	e Divided by Free-Flow	Travel Time (Congestion	ı Index)
	(1)	(2)	(3)	(4)
		1. Morn	ing Peak	
Strike	0.117*	0.134**	0.123*	0.123*
[1.47]	(0.052)	(0.042)	(0.049)	(0.051)
N P ²	2,454	2,454	2,454	2,454
R^2	0.392	0.539	0.621	0.630
		2. Eveni	ing Peak	
Strike	0.056**	0.065^{*}	0.064*	0.062
[1.49]	(0.020)	(0.024)	(0.027)	(0.030)
Ν	2,454	2,454	2,454	2,454
R^2	0.291	0.351	0.473	0.482
		2 411	Doola	
0. 1	0.00.0**	J. All		0.002*
Strike	(0.086^{**})	0.099**	0.094**	0.093*
[1.+0] N	(0.051)	(0.020)	(0.055)	(0.050)
R^2	2,454	2,454	2,454	2,454
	0.025	0.17.1	0.000	0.071
		4. Al	l Day	
Strike	0.050**	0.056**	0.051**	0.050**
[1.30]	(0.013)	(0.013)	(0.016)	(0.017)
N_{-2}	2,454	2,454	2,454	2,454
R^2	0.306	0.397	0.533	0.547
		5. City Stree	ets – All Day	
Strike	0.047***	0.054***	0.048***	0.048***
[1.36]	(0.007)	(0.011)	(0.009)	(0.010)
Ν	2,454	2,454	2,454	2,454
R^2	0.350	0.478	0.591	0.601
		6 Freeway	ve – All Dav	
Stuiles	0.050*	0.055**	0.051*	0.040
[1.25]	(0.030)	(0.019)	(0.022)	(0.024)
N N	2 454	2 454	2 454	2 454
R^2	0.252	0.308	0.503	0.522
City FE	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes
Controls		Yes	Yes	Yes
City×Time City×Weather			Yes	Yes
				108

TABLE 3: The Effect of Strikes on Travel Times

NOTES: Number of one-day strikes used in estimation sample: 26. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include city fixed effects. Time FE include day-of-week, week-of-year, and year. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City×Time are interactions of city indicators with all Time FE. City×Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variabl	es: Number of Vehicl	e Crashes and Accident-	Related Injuries	
	(1)	(2)	(3)	(4)
		1. Vehicle Crashe	s – Morning Peak	
Strike	0.616*	0.618*	0.616*	0.607*
[4.280]	(0.254)	(0.275)	(0.259)	(0.250)
$N = p^2$	12,253	12,238	12,238	12,238
<i>K</i> -	0.405	0.428	0.472	0.478
		2. Slightly Injured Pe	rsons – Morning Peak	
Strike	0.761***	0.765**	0.793**	0.790**
[3.940]	(0.151)	(0.181)	(0.201)	(0.192)
Ν	12,253	12,238	12,238	12,238
R^2	0.388	0.408	0.449	0.455
	3 Se	riously or Fatally Inju	ed Persons _ Morning	Peak
C to:l.o	0.011			0.012
Strike [0 354]	-0.011	-0.014	-0.012	-0.013
[0.551] M	(0.059)	(0.037)	(0.055)	(0.055)
$\frac{N}{R^2}$	12,253	12,238	12,238	12,238
A	0.070	0.101	0.124	0.127
		4. Vehicle Crashe	es – Evening Peak	
Strike	0.269	0.284	0.080	0.0836
[6.962]	(0.418)	(0.481)	(0.476)	(0.475)
N	12,253	12,238	12,238	12,238
R^2	0.540	0.561	0.599	0.607
		5. Slightly Injured Pe	rsons – Evening Peak	
Strike	0.546	0.561	0.357	0.388
[6.786]	(0.463)	(0.517)	(0.533)	(0.527)
Ν	12,253	12,238	12,238	12,238
R^2	0.497	0.514	0.552	0.558
	6 54	ariously or Fatally Injug	rad Parsons Evaning	Pook
C t :: l	0.100			0.002
[0 648]	-0.106	-0.101	-0.090	-0.093
[0.0+0]	(0.055)	(0.030)	(0.055)	(0.055)
R^2	0.155	12,238	12,238	12,238
21	0.135	0.102	0.107	0.172
Monitor FE	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes
Controls		Yes	Yes	Yes
City × 11me City × Weather			res	res Ves
				105

TABLE 4: The Effect of Strikes on Vehicle Crashes and Accident-Related Injuries

NOTES: Number of one-day strikes used in estimation sample: 71. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include city fixed effects. Time FE include day-of-week, week-of-year, and year. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City×Time are interactions of city indicators with all Time FE. City×Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variab	le: Mean Hourly Pollu	ition Emissions in $\mu g/m$	3	
	(1)	(2)	(3)	(4)
		1. PM10 – M	Iorning Peak	
Strike	5.150**	5.013**	5.566**	5.334**
[37.64]	(1.328)	(1.600)	(1.607)	(1.653)
$\frac{N}{R^2}$	33,049	33,007	33,007	33,007
	0.184	0.313	0.342	0.351
		2. NO2 – M	orning Peak	
Strike	2.749	2.840	3.277*	3.314*
[76.85]	(1.433)	(1.460)	(1.417)	(1.427)
$\frac{N}{R^2}$	38,586	38,525	38,525	38,525
	0.398	0.490	0.510	0.519
		3. PM10 – E	vening Peak	
Strike	1.085	0.547	0.464	0.292
[35.30]	(2.394)	(2.677)	(2.942)	(2.940)
$\frac{N}{R^2}$	33,778	33,737	33,737	33,737
	0.196	0.305	0.338	0.350
		4. Mean NO2 -	- Evening Peak	
Strike	-0.487	-0.464	-0.973	-1.063
[77.20]	(2.473)	(3.073)	(3.298)	(3.436)
$\frac{N}{R^2}$	39,528	39,468	39,468	39,468
	0.347	0.436	0.463	0.478
Monitor FE Time FE Controls City×Time City×Weather	Yes Yes	Yes Yes Yes	Yes Yes Yes Yes	Yes Yes Yes Yes Yes

TABLE 5: The Effect of Strikes on Particle Pollution and Nitrogen Dioxide Emissions

NOTES: Number of one-day strikes used in estimation sample: 68. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include monitor fixed effects. Time FE include day-of-week, week-of-year, and year. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City \times Time are interactions of city indicators with all Time FE. City \times Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variable	e: Mean Hourly Pollu	tion Emissions in $\mu g/m$	3	
	(1)	(2)	(3)	(4)
		1. SO2 – Me	orning Peak	
Strike [6.47]	0.361 (0.385)	0.234 (0.380)	0.089 (0.255)	0.186 (0.190)
$\frac{N}{R^2}$	14,068 0.187	14,038 0.227	14,038 0.272	14,038 0.297
		2. SO2 – Ev	ening Peak	
Strike [5.03]	-0.040 (0.220)	-0.275 (0.297)	-0.238 (0.222)	-0.184 (0.235)
$\frac{N}{R^2}$	14,377 0.259	14,349 0.300	14,349 0.361	14,349 0.371
Monitor FE Time FE Controls City×Time City×Weather	Yes Yes	Yes Yes Yes	Yes Yes Yes Yes	Yes Yes Yes Yes Yes

TABLE 6: The Effect of Strikes on Placebo Air Pollution

NOTES: Number of one-day strikes used in estimation sample: 45. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include monitor fixed effects. Time FE include day-of-week, week-of-year, and year. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City \times Time are interactions of city indicators with all Time FE. City \times Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variable: Number of Hospitalized Patients per Day				
	(1)	(2)	(3)	
	Full sample	Ages below 5	Ages 65 and above	
	1. Respirato	ory Diseases (ICD-10 c	odes J00-J99)	
Strike	0.963	0.879**	0.145	
	(1.746)	(0.208)	(0.829)	
Ν	11,000	11,000	11,000	
R^2	0.924	0.692	0.861	
[Mean]	[61.09]	[7.82]	[22.09]	
2. Abnormalities of Breathing (ICD-10 code R06)				
Strike	0.160**	0.074**	0.049	
	(0.048)	(0.018)	(0.096)	
Ν	11,000	11,000	11,000	
R^2	0.182	0.098	0.089	
[Mean]	[1.27]	[0.22]	[0.39]	
City FE	Yes	Yes	Yes	
Time FE	Yes	Yes	Yes	
Controls	Yes	Yes	Yes	
City ×Time	Yes	Yes	Yes	
City×Weather	Yes	Yes	Yes	

TABLE 7. The Effect of Strikes on Hospitalizations	TABLE 7:	The E	Effect of	Strikes on	Hospitalizations	
--	----------	-------	-----------	------------	------------------	--

NOTES: Number of one-day strikes used in estimation sample: 57. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include city fixed effects. Time FE include day-of-week, week-of-year, and year. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City×Time are interactions of city indicators with all Time FE. City×Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variab	ble: Hourly Passenger	Vehicle Flows per Monit	or	
	(1)	(2)	(3)	(4)
		1. Freeways –	Morning Peak	
Strike	160.5***	136.5***	128.4***	131.6***
[5,239]	(26.47)	(10.75)	(13.79)	(12.09)
Mass event	-29.86	-40.48*	-6.275	-0.234
	(26.40)	(15.41)	(11.72)	(10.09)
$\frac{N}{R^2}$	213,160	212,896	212,896	212,896
	0.899	0.903	0.921	0.922
		2. Federal Roads	– Morning Peak	
Strike	63.05**	57.29**	72.42***	77.67***
[1,790]	(12.77)	(15.27)	(8.021)	(9.383)
Mass event	4.891	-3.430	2.691	-4.469
	(15.19)	(4.688)	(3.764)	(6.443)
$\frac{N}{R^2}$	102,704	102,540	102,540	102,540
	0.921	0.924	0.961	0.962
		3. Freeways –	Evening Peak	
Strike	125.2**	103.2***	88.39**	91.51***
[5,785]	(28.54)	(21.37)	(19.79)	(17.91)
Mass event	-20.25	-25.85	-10.23	-3.999
	(24.90)	(17.89)	(7.039)	(7.196)
$\frac{N}{R^2}$	213,160	212,896	212,896	212,896
	0.937	0.939	0.950	0.950
		4. Federal Road	s – Evening Peak	
Strike	21.81	13.95	26.28	37.82**
[2,121]	(17.22)	(18.02)	(14.31)	(10.27)
Mass event	14.82	9.474	-0.917	-5.533
	(20.05)	(12.21)	(3.856)	(5.104)
$\frac{N}{R^2}$	102,704	102,540	102,540	102,540
	0.960	0.962	0.972	0.973
Monitor FE	Yes	Yes	Yes	Yes
Controls City×Time City×Weather	168	Yes	Yes Yes	Yes Yes Yes

TABLE 8: The Effect of Public Transit Strikes on Car Traffic – Controlling for Mass Events

NOTES: Number of one-day strikes used in estimation sample: 64 for freeways, 45 for federal roads. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include monitor fixed effects. Time FE include hour-of-day, day-of-week, hour-of-day×day-of-week, week-of-year, year and holiday fixed effects. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City×Time are interactions of city indicators with all Time FE. City×Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Dependent Variable:	Hourly Passenger Ve	hicle Flows per Monitor		
	(1)	(2)	(3)	(4)
	Freeways	Federal Roads	Freeways	Federal Roads
	(morning peak)	(morning peak)	(evening peak)	(evening peak)
One-day strike	131.6***	77.77***	91.35**	37.83**
	(12.12)	(9.528)	(17.98)	(10.14)
Multi-day strike	113.7**	54.24**	94.35**	-0.619
	(33.57)	(12.58)	(23.62)	(27.36)
$\frac{N}{R^2}$	213,892	103,128	213,892	103,128
	0.922	0.962	0.950	0.973
[Mean]	[5,239]	[1,790]	[5,785]	[2,121]
Time FE	Yes	Yes	Yes	Yes
Weather	Yes	Yes	Yes	Yes
City×Time	Yes	Yes	Yes	Yes
City×Weather	Yes	Yes	Yes	Yes

TABLE 9: The Effect of One-Day and Multi-Day Strikes on Car Traffic

NOTES: Number of one-day strikes used in estimation sample: 64 for freeways, 45 for federal roads. Multi-day strikes used in estimation sample include 12 events covering 41 city-day observations for freeways and 10 events covering 37 city-day observations for federal roads. Mean of the dependent variable on strike-free days reported in square brackets. All regressions include monitor fixed effects. Time FE include hour-of-day, day-of-week, hour-of-day×day-of-week, week-of-year, year and holiday fixed effects. Controls include a dummy for school holidays and the following weather variables: atmospheric temperature, amount of precipitation, wind speed, and a snow cover dummy. City×Time are interactions of city indicators with all Time FE. City×Weather are interactions of city indicators with all weather variables. Weights are the number of observations per station over the number of observations per city. Cluster-robust standard errors in parentheses. * 10%, **5%, *** 1% confidence level.

Appendix for Online Publication



FIGURE A1: Location of Traffic and Air Monitors – Berlin

NOTES: Triangles indicate traffic monitors on freeways, diamonds indicate traffic monitors on federal roads, stars indication air monitors. Map tiles by Stamen Design, under CC BY 3.0. Data by OpenStreeMap, under CC BY SA. Scale 1:250,000.



FIGURE A2: Location of Traffic and Air Monitors – Hamburg

NOTES: Triangles indicate traffic monitors on freeways, diamonds indicate traffic monitors on federal roads, stars indication air monitors. Map tiles by Stamen Design, under CC BY 3.0. Data by OpenStreeMap, under CC BY SA. Scale 1:200,000.



FIGURE A3: Location of Traffic and Air Monitors – Munich

NOTES: Triangles indicate traffic monitors on freeways, diamonds indicate traffic monitors on federal roads, stars indication air monitors. Map tiles by Stamen Design, under CC BY 3.0. Data by OpenStreeMap, under CC BY SA. Scale 1:200,000.



FIGURE A4: Location of Traffic and Air Monitors – Cologne

NOTES: Triangles indicate traffic monitors on freeways, diamonds indicate traffic monitors on federal roads, stars indication air monitors. Map tiles by Stamen Design, under CC BY 3.0. Data by OpenStreeMap, under CC BY SA. Scale 1:200,000.



FIGURE A5: Location of Traffic and Air Monitors - Frankfurt

NOTES: Triangles indicate traffic monitors on freeways, diamonds indicate traffic monitors on federal roads, stars indication air monitors. Map tiles by Stamen Design, under CC BY 3.0. Data by OpenStreeMap, under CC BY SA. Scale 1:200,000.



FIGURE A6: Passenger Vehicle Flows over the Course of an Average Weekday



FIGURE A7: Air Pollution over the Course of an Average Weekday

Year	Berlin	Cologne	Frankfurt	Hamburg	Munich
2002		05/27/2002 (4:00-8:00)			05/27/2002 (4:00- 8:00)
		12/17/2002 (4:00-7:30)			12/16/2002 (all day)
2003		03/06/2003 (6:00-6:45)	03/06/2003 (6:00-6:45)	03/06/2003 (6:00-6:45)	03/06/2003 (6:00-6:45)
2004	04/21/2004 (7:00-8:00)				
	04/23/2004 (18:00-20:00)				
2005	05/24/2005 (3:30-10:00)				09/15/2005 (all day)
2006		09/29/2006 (4:00-6:00)			
	07/03/2007 (5:00-9:00)	07/03/2007 (5:00- 9:00)	07/03/2007 (5:00-9:00)	07/03/2007 (5:00-9:00)	07/03/2007 (5:00- 9:00)
	07/10/2007 (8:00-10:15)	07/10/2007 (8:00-10:15)		07/10/2007 (8:00-10:15)	07/10/2007 (8:00-10:15)
2007	08/09/2007 (8:00-10:00)			08/09/2007 (8:00-10:00)	
	10/05/2007 (8:00-11:00)	10/05/2007 (8:00- 11:00)	10/05/2007 (8:00-11:00)	10/05/2007 (8:00-11:00)	10/05/2007 (8:00-11:00)
	10/12/2007 (all day)	10/12/2007 (all day)	10/12/2007 (all day)	10/12/2007 (all day)	10/12/2007 (all day)
	10/18/2007 (2:00-11:00)	10/18/2007 (2:00-11:00)	10/18/2007 (2:00-11:00)	10/18/2007 (2:00-11:00)	10/18/2007 (2:00-11:00)
2008		02/22/2008 (4:00-12:00)	02/22/2008 (3:00-7:30)		
2009		01/29/2009 (6:30-9:00)	02/25/2009 (all day)		02/03/2009 (3:30-15:30)
2007					02/27/2009 (all day)
	02/09/2010 (3:00-14:00)	02/04/2010 (3:00-6:30)	02/01/2010 (all day)	01/20/2010 (all day)	09/10/2010 (4:00-10:00)
2010		10/26/2010 (4:00-9:00)	02/05/2010 (all day)	01/29/2010 (all day)	09/15/2010 (all day)
			10/26/2010 (5:00-8:30)	02/18/2010 (<i>3:00-15:00</i>)	10/26/2010 (4:00-19:00)
	02/22/2011 (6:00-8:00)	02/22/2011 (6:00-8:00)	02/22/2011 (6:00-8:00)	02/22/2011 (6:00-8:00)	02/22/2011 (6:00-8:00)
2011	03/10/2011 (<i>4:00-10:00</i>)	02/25/2011 (8:30-11:30)	02/25/2011 (8:30-11:30)	02/25/2011 (8:30-11:30)	02/25/2011 (8:30-11:30)
		03/10/2011 (4:00-10:00)	03/10/2011 (<i>4:00-10:00</i>)	03/10/2011 (<i>4:00-10:00</i>)	03/10/2011 (4:00-10:00)

TABLE A1: Public Transit Strikes of One Day or Less (2002-2011)

NOTES: Table lists dates and duration of one-day strikes in public transportation during the period 2002-2011. One-day strikes labeled "all day" affected the entire operating hours of the services in question.

TABLE A	2: Summe	ary Statistics
		-

	Ν	Mean	Std dev	Min	Max
	Panel A: Car Traffic				
# Freeway cars per hour (morning peak)	213,160	5,240	2,291	113	12,911
# Freeway cars per hour (evening peak)	213,160	5,786	2,253	0	13,142
# Federal road cars per hour (morning peak)	102,704	1,789	979	61	5,039
# Federal road cars per hour (evening peak)	102,704	2,120	1,069	307	5,463
	Panel B: Congestion				
Congestion Index (morning peak)	2,454	1.47	0.20	1.04	3.03
Congestion Index (evening peak)	2,454	1.49	0.20	1.13	3.36
Congestion Index (all peaks)	2,454	1.48	0.16	1.09	2.63
Congestion Index (all day)	2,454	1.31	0.09	1.09	2.04
Congestion Index (city streets - all day)	2,454	1.36	0.07	1.18	2.01
Congestion Index (highways - all day)	2,454	1.25	0.11	1.04	2.11
	Panel C: Accidents				
# Vehicle crashes (morning peak)	12,253	4.28	3.28	0	27
# Vehicle crashes (evening peak)	12,253	6.96	4.82	0	39
<pre># Slightly injured (morning peak)</pre>	12,253	3.94	3.47	0	26
<pre># Slightly injured (evening peak)</pre>	12,253	6.78	5.27	0	41
# Seriously or fatally injured (morning peak)	12,253	0.35	0.66	0	6
# Seriously or fatally injured (evening peak)	12,253	0.65	0.95	0	8
	Panel D: Pollution				
Mean PM10 in $\mu g/m^3$ (morning peak)	33,049	37.68	21.28	2	463
Mean PM10 $\mu g/m^3$ (evening peak)	33,778	35.32	20.30	1	273
Mean NO2 in $\mu g/m^3$ (morning peak)	38,586	76.87	29.72	2	257
Mean NO2 in $\mu g/m^3$ (evening peak)	39,528	77.22	31.79	5	350
Mean SO2 in $\mu g/m^3$ (morning peak)	14,068	6.46	6.41	0	101
Mean SO2 in $\mu g/m^3$ (evening peak)	14,377	5.03	4.58	0	70
	Panel E: Hospitalizations				
# Respiratory (all patients)	11,015	61.08	36.02	3	250
# Respiratory (ages below 5)	11,015	7.82	5.76	0	45
# Respiratory (ages 65 and above)	11,015	22.08	15.05	0	112
# Breathing (all patients)	11,015	1.27	1.23	0	8
# Breathing (ages below 5)	11,015	0.22	0.49	0	5
# Breathing (ages 65 and above)	11,015	0.39	0.65	0	4
	Panel F: Control Variables				
Mean Temperature (°C)	12,253	10.42	7.58	-15	30
Precipitation (mm)	12,238	3.41	1.58	0	14
Wind speed (m/s)	12,253	1.96	4.61	0	130
Snow cover	12,253	0.07	0.25	0	1
School vacations	12,253	0.26	0.44	0	1

NOTES: Table lists descriptive statistics (number of observations, mean, standard deviation, minimum, and maximum) of all variables in the data set. The data summarized in Panel A are based on monitor-hour observations. The data summarized in Panel B are based on monitor-day observations. The data summarized in Panels B, C, E, F are based on city-day observations.